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Received Strips
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Performed Tests:

• IV @ TROOM

• IV @ TLN

• Thermal Stresses
• DCR
• Extended IV (up to VBD + 10V)



First Box – 2400 SiPMs
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ROOM LN

Problem in Vbias  Electrical Connection
→ Fixed after changing the cable



Problems on VBIAS distribution
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Not «visible» by looking at data: Rq is higher than «normal» but no hint on problems 

Reverse is normal
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Visible by looking at data in forward polarization
Reverse polarization is normal

Fixed by changing cables



DCR - examples

• CACTUS setup is near to INFN mechanical workshop

• Some channels are noisy (here below one of them)

• NOT due to SiPMs with high dark current:
• Checked by applying 0 V to the SiPMs and by changing SiPMs positions
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My SQL Database
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Example on extracted data
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Conclusions

• CACTUS @Bologna working well:
• 120 SiPMs /day
• First box completed;
• Second box (50% completed)

• To be completed by the end of this month
• Third box received
• We had some problems in the Vbias distribuition → solved
• We experience some noise in the DCR measurements

• Extended IV doesn’t show problematic sipms

• Local DB created in order to:
• Backup data;
• Create .csv files to push into DUNE DB
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